Author’s response to reviews

Title: Volume, patterns, and types of sedentary behavior and cardio-metabolic health in children and adolescents: a cross-sectional study

Authors:

Valerie Carson (8vc9@queensu.ca)
Ian Janssen (ian.janssen@queensu.ca)

Version: 5 Date: 18 April 2011

Author’s response to reviews:

Formatting changes (references, tables, highlighting) as requested.